Author's response to reviews

Title: Large aortic aneurysm mimicking a cardiac tumor

Authors:

Zhen-Yu Liao (ariex327@mail2000.com.tw)
Jui-Peng Tsai (amanda1015h2o@yahoo.com.tw)
Jen-Yuan Kuo (jykuo@ms1.mmh.org.tw)
Chung-Lieh Hung (jotaro3791@gmail.com)

Version: 3 Date: 9 August 2010

Author's response to reviews:

Dear Sir,

Please kindly find this revised manuscript entitled “Large aortic aneurysm mimicking a cardiac tumor” that we are submitting for possible publication in “Cardiovascular Ultrasound”. Thank you for the opportunity to further revise our manuscript entitled “Large aortic aneurysm mimicking a cardiac tumor.” We appreciate the expert comments of the reviewers and editors, and we have responded in a point by point fashion to these comments and have made appropriate changes in the manuscript as suggested. In addition, we have since then made the formatting changes as requested.

Please allow me to assure you that the information within this manuscript has not previously been submitted for publication nor has it been published in whole or in part elsewhere. As a first author, I would like to thank Dr. Hung, Chung-Lieh, the corresponding author, for editing and reviewing the manuscript. I am also willing to make any additional change to this manuscript if any in the future, as recommended by the reviewers. I have confidence in your judgment and recommendations. By submitting this manuscript for possible publication, I acknowledge that the copyright for the enclosed material is transferred to the journal.

The manuscript, as submitted or its essence in another version, is not under consideration for publication elsewhere, and will not be published elsewhere while under consideration by “Cardiovascular Ultrasound”. All authors have made substantial contributions to the case report, and that all authors have read and endorse the manuscript and conclusions.

Sincerely